Author's response to reviews

Title: Pulmonary stenosis development and reduction of pulmonary arterial hypertension in atrioventricular septal defect

Authors:
	emeline barth (emelinebarth@yahoo.fr)
hélène bouvaist (HBouvaist@chu-grenoble.fr)
gérard ninet (gerard.ninet@chu-lyon.fr)
gérald vanzetto (GVanzetto@chu-grenoble.fr)
stéphanie marlière (SMarliere@chu-grenoble.fr)

Version: 4 Date: 30 August 2009

Author’s response to reviews:

COVER LETTER:
- Author to whom address correspondence: Emeline Barth
  0033 6 88 11 00 89 (France)
  mail: emelinebarth@yahoo.fr

- No previous or concurrent submissions for this case report
- 1643 words counting the title page
- Last revisions made as answering reviewers:
- Treatments until surgery: digitalo-diuretics which have been stopped in 1987.
- Surgeon closed the atrial and ventricular parts of the AVSD
- Vasoreactivity test of pulmonary arterial hypertension reversibility: Notric oxide
- Figure 3: Proximal and distal pulmonary artery diameter = 28 mm